Author’s response to reviews

Title: Mobile Phone Intervention for Increasing Adherence to Treatment for Type 2 Diabetes in an Urban Area of Bangladesh: Protocol for a Randomized Controlled Trial

Authors:

Sheikh Mohammed Shariful Islam (shariful.islam@icddrb.org)
Andreas Lechner (andreas.lechner@med.uni-muenchen.de)
Uta Ferrari (uta.ferrari@med.uni-muenchen.de)
Dewan S Alam (dsalam@icddrb.org)
Rolf Holle (holle@helmholtz-muenchen.de)
Jochen Seissler (jochen.seissler@med.uni-muenchen.de)
Louis W Niessen (niessen@icddrb.org)

Version: 2 Date: 20 September 2013

Author’s response to reviews:

The manuscript has been revised for language mistakes and corrections made. We have also included one more co-author whom we could not include earlier as we did not get his signature for approval of the manuscript.